Improving Menstrual Health Throughout the Reproductive Life Course - Original Research Article

The role of probiotics in improving menstrual health in women with primary dysmenorrhoea: A randomized, double-blind, placebo-controlled trial (the PERIOD study)



Women's Health Volume 20: 1-11 © The Author(s) 2024 Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/17455057241234524 journals.sagepub.com/home/whe



Izyan Atiqah Zakaria¹, Nur Aini Mohammed Zain¹, Chew Kah Teik¹, Muhammad Azrai Abu¹, Ani Amelia Zainuddin¹, Nor Haslinda Abdul Aziz¹, Nazarudin Safian², Norfilza Mohd Mokhtar³, Raja Affendi Raja Ali⁴, Ng Beng Kwang¹, Nor Azlin Mohamed Ismail¹, Muhammad Rafiuddin Hamizan¹, Wira Sorfan Ab Razak¹ and Abdul Ghani Nur Azurah¹

Abstract

Background: Primary dysmenorrhea is associated with poorer quality of life; however, the causal mechanism remains unclear. A vast body of literature supports the use of oral probiotics for relief from the symptoms of endometriosis; however, to our knowledge, no study has prescribed probiotics for primary dysmenorrhea.

Objective: The aim of this study is to investigate the effects of 3-month supplementation with oral probiotics on quality of life and inflammatory markers in women with primary dysmenorrhea.

Design: Randomized placebo-controlled trial.

Methods: A total of 72 patients (36 patients in each arm) were randomized to receive either oral sachets containing 5 billion colony-forming units each of Lactobacillus acidophilus BCMC (BCrobes Microbial Cells) 12130, Lactobacillus casei subsp BCMC 12313, Lactobacillus lactis BCMC 12451, Bifidobacterium bifidum BCMC 02290, Bifidobacterium longum BCMC 02120, and Bifidobacterium infantis BCMC 02129 each or placebo twice daily for 3 months. Main outcome measures were visual analog scale, verbal rating scale, physical and mental health scores using Short-Form 12-Item version 2 questionnaire, frequency of nonsteroidal anti-inflammatory drug use, and changes in inflammatory markers (interleukin-6, interleukin-8, and tumor necrosis factor alpha) before and after treatment.

Results: There was no significant difference in the quality of life scores between the probiotic and placebo groups. Both groups showed significant improvement in pain (visual analog scale) and severity (verbal rating scale) scores but the probiotic group had much lower nonsteroidal anti-inflammatory drug use (odds ratio: 0.69, 95% confidence interval: 0.26–1.83) and better mental health scores (mean change: 6.5, p=0.03 versus 6.1, p=0.08) than the placebo group. There was a significant confounding effect of nonsteroidal anti-inflammatory drug use on quality of life scores. No significant difference was found in inflammatory cytokines.

¹Department of Obstetrics and Gynaecology, Faculty of Medicine, National University of Malaysia, Kuala Lumpur, Malaysia ²Department of Community Health, Faculty of Medicine, National

University of Malaysia, Kuala Lumpur, Malaysia

³Department of Physiology, Faculty of Medicine, National University of Malaysia, Kuala Lumpur, Malaysia

⁴Department of Medicine, Faculty of Medicine, National University of Malaysia, Kuala Lumpur, Malaysia

Corresponding author:

Abdul Ghani Nur Azurah, Department of Obstetrics and Gynaecology, Faculty of Medicine, Jalan Yaakob Latif, 56100 Kuala Lumpur National University of Malaysia, Kuala Lumpur, Malaysia. Email: azurah@ppukm.ukm.edu.my

 (\mathbf{i}) Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the SAGE and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).



Conclusion: Tested oral probiotics improved mental health and potentially reduced the use of nonsteroidal antiinflammatory drugs; however, there was no significant change in inflammatory markers. Further research with a larger sample size is needed to confirm the findings.

Registration: This study is registered under ClinicalTrials.gov (NCT04119011).

Plain Language Summary

Use of Probiotic in Primary Dysmenorrhoea

This study looked at whether taking probiotics (good bacteria) for 3 months could improve the quality of life and reduce pain in women with painful periods. The study found that probiotics did not significantly improve quality of life scores, but did reduce the use of painkillers and improve mental health scores. However, the probiotics did not have a significant effect on inflammatory markers in the body. More research is needed to confirm the findings.

Keywords

inflammatory markers, menstrual disorders, primary dysmenorrheal, probiotics, quality of life

Date received: 22 June 2023; revised: 19 January 2024; accepted: 6 February 2024

Introduction

Primary dysmenorrhea is a common problem faced by women of reproductive age and is often underappreciated. Women with dysmenorrhea often have limited work and school functioning. It is a major cause of absenteeism and reduced quality of life.¹⁻³ Young women with dysmenorrhea are also at risk of chronic pain later in life owing to central sensitization.⁴ Primary dysmenorrhea affects women with regular menstrual cycles in the absence of organic diseases such as endometriosis, uterine leiomvoma, adenomvosis, or other uterine or ovarian pathologies.^{5,6} However, the exact pathogenesis of primary dysmenorrhea remains unknown. Many studies have attributed pain, uterine contraction, and inflammation to prostaglandins (PGs).7 The intensity of menstrual cramps and associated symptoms of dysmenorrhea are directly proportional to the amount of PGF2 α released.⁸ The synthesis of PGF2 α (Prostaglandin F 2 alpha) is proven to be upregulated by proinflammatory cytokines, such as interleukins (IL)-6 and IL-8, and downregulated by anti-inflammatory markers like IL-11.9 Leukotrienes, another by-product of the lipoxygenase enzyme pathway, also exacerbate uterine contractions; however, their actions are not affected by nonsteroidal anti-inflammatory drugs (NSAIDs).¹⁰ This may explain why NSAIDs are not effective in some patients with primary dysmenorrhea. The effect of other mediators, such as progesterone, vasopressin, and calcium channels, has also been examined in some studies; however, evidence regarding the usefulness of their therapeutic action is still lacking.11

The conventional treatment for primary dysmenorrhea relies on the use of oral NSAIDs and hormonal pills to reduce inflammation and suppress ovulation. However, these treatments have been found to have adverse effects on health and cancer risk, particularly with long-term use. NSAIDs can lead to organ damage via the build-up of oxidative stress species, while oral contraceptive pill users have an increased risk of breast cancer, as reported in recent literature.^{12,13} Furthermore, some patients fail to respond to NSAIDs, and hormonal therapy is contraindicated in some cases; therefore, limited treatment options are available.¹⁴ The search for an alternative treatment with few side effects, especially for long-term use, is still ongoing.

Probiotics can be used as an alternative treatment option for primary dysmenorrhea. Probiotic supplements have been shown to colonize the human intestine and confer many health benefits, particularly in immunomodulatory and inflammatory conditions.¹⁵ Probiotics are live microorganisms that have beneficial health effects in many disease states. Certain metabolic processes are regulated through diverse human microbiomes that promote favorable conditions in the host's intestinal environment.¹⁶ In patients with endometriosis, the use of Lactobacillus gasseri GG has been found to suppress the development of ectopic endometriotic lesions in a murine model and lead to improved visual analog scale (VAS) and verbal rating scores for dysmenorrhea.¹⁷ Khodaverdi et al.¹⁸ reported improved pain scores after the use of oral Lactobacillus supplements in patients with endometriosis.

However, studies on the use of probiotics in women with primary dysmenorrhea are limited. Hence, we designed this study to examine the potential benefits of probiotics as a novel treatment for women with primary dysmenorrhea. We hypothesized that oral supplementation with probiotics for 3 months will restore gut dysbiosis and reduce inflammatory cytokines, thus improving the quality of life in women with primary dysmenorrhea.

Materials and methods

Participants

A sample size of 30 was calculated based on a study by Safdari and Parvin to detect a mean score difference of 0.92 in the VAS scores between the probiotic and placebo groups, with a power of 80% and a two-sided type-1 error of 0.05.¹⁹ Considering a dropout rate of 20%, the final sample size of 72, with 36 participants in each arm, was determined. Due to inadequate funding, only 28 samples were analyzed for the analysis of inflammatory markers. The pretreatment and post-treatment samples were randomly selected from each treatment group (eight from the probiotic group and six from the placebo group).

Women who presented with primary dysmenorrhea at gynecology clinics or wards of the University of Kebangsaan Malaysia Medical Center (UKMMC) were eligible to participate in the study. All the participants provided written informed consent. Premenopausal women with a regular menstrual cycle, primary dysmenorrhea, and compliant with oral sachets consumed twice daily for 3 months were eligible. Primary dysmenorrhea was diagnosed according to the history and medical examination data (recurrent menstrual pain on the first 2 days of menstruation without identifiable pathology), confirmed by pelvic ultrasonography when the diagnosis was doubtful. Women on intrauterine copper devices, with a history of hormonal treatment or contraceptives within the past 3 months, who had food allergies or were lactose intolerance, who had suspected or confirmed tumor or malignancy, who frequently used laxative or antidiarrheal drugs, or who were receiving treatment for allergic diseases were excluded.

Recruitment

The researchers identified eligible participants from among the patients at the clinic and/or ward. Women who agreed to participate and met the inclusion criteria were counseled regarding the study, and written consent was obtained. Seventy-eight potential candidates were identified between October 2019 and December 2019. Six women were excluded because they were unwilling to participate or did not meet the inclusion criteria. A total of 72 women were randomized to placebo and probiotic groups. At the end of the study, five participants were lost to follow-up. The final analysis included 67 (93%) of the randomized women, as shown in Figure 1. This study adhered to the CONSORT checklist.

Study procedure

This double-blind, randomized, placebo-controlled trial was conducted at the UKMMC between October 2019 and March 2020. Upon recruitment, the participants' demographic data and menstrual history were collected. The participants were asked to complete the VAS, verbal rating scale (VRS), and Short-Form 12-item version 2 (SF12v2) questionnaires on the first 2 days of their menses. Blood samples were collected on day 2 of menstruation in a 5-mL plain tube. The samples were centrifuged at 4000 r for 20 min to separate the serum that was subsequently stored at -80° C until further processing for inflammatory markers. All the participants were given a pain diary to document menstrual flow, pain intensity, and frequency of oral NSAID use throughout the 3-month treatment period. The use of analgesics other than NSAIDs was not assessed. The patients were also asked to maintain a pain diary.

The participants were randomized upon recruitment using a computerized randomization sequence in a 1:1 ratio, and designated A or B, and treatments were dispensed accordingly. Probiotic and placebo sachets were prepared by a probiotic-manufacturing company using identical packaging. Each sachet was individually labeled as A or B and contained either probiotics or placebo. The sachets were then packaged into identical tamper-proof boxes with details regarding the expiry date, storage instructions, and instructions for use. Only the manufacturer was aware of the contents of the sachets. Both the researchers and participants were blinded to the study grouping until data collection was completed.

The treatment group received oral probiotic sachets containing 5 billion colony-forming units each of Lactobacillus acidophilus BCMC 12130, Lactobacillus casei subsp BCMC 12313, Lactobacillus lactis BCMC 12451. *Bifidobacterium* bifidum BCMC 02290. Bifidobacterium longum BCMC 02120. and Bifidobacterium infantis BCMC 02129, while the placebo group received identical sachets containing excipients, namely maltodextrin, lactose, oligosaccharides, citric acid, ascorbic acids, and flavoring agents (B-Crobes, Subang Java, Malaysia). The participants were instructed to consume one sachet twice daily with meals for 3 months. The participants received 250 mg of oral mefenamic acid (Ponstan) for use when needed during the study period.

The participants were contacted once a month, either by face-to-face consultation or via mobile phone, to assess and emphasize compliance and monitor side effects. They were deemed compliant if they consumed more than 80% of treatment sachets, as determined by counting the remaining sachets at each visit. At the end of the third month of treatment, the participants were asked to assess the first 2 days of their menses, whereby the VAS, VRS, and SF12v2 questionnaires and blood collection were repeated. The patients were also checked.

Outcomes

The primary outcome was improvement in quality of life, reflected by the pain, severity, physical health, and mental health scores and frequency of NSAID use on the second day of menstruation before and after treatment. The pain score was measured using the 10-point VAS and severity score, using the VRS (grades 0–3).^{20–22} The SF12v2 questionnaire assessed the impact of cyclical menstrual pain on



Figure 1. Flowchart showing numbers of participants at each stage of the trial.

physical and mental health, where the score was transformed into a score out of 100% according to the given formula.^{20–22} Permission for the use of this questionnaire was obtained from the primary author.

The secondary outcome was the difference in the concentrations of inflammatory markers before and after treatment. The levels of inflammatory markers IL-6, IL-8, and tumor necrosis factor alpha (TNF- α) were measured using the multiplex immunoassay MILLIPLEX MAP Human High Sensitivity T-Cell Magnetic Bead Panel (Prima Nexus, Kuala Lumpur, Malaysia). Each sample was incubated with cytokine antibodies, washed, and analyzed using a Magpix reader (Luminex Corp., Austin, TX, USA). The fluorescence unit was then converted to a cytokine concentration unit to obtain the measurement levels. In 2013, Ma et al.⁹ investigated the changes in inflammatory cytokines during menstrual cycles in a small number of participants (six with dysmenorrhea and three controls). Similarly, owing to limited funding, only 28 samples were randomly selected and analyzed in our study (eight and six pretreatment and post-treatment samples from the probiotic and placebo groups, respectively).

Statistical analysis

All analyses were performed using the Statistical Package for Social Sciences (SPSS), version 23. Descriptive analysis was used to report the baseline characteristics of the Table 1. Baseline characteristics of participants.

Characteristics	Probiotic $(n=36)$	Placebo (n=36)	Þ
Participant characteristics			
Age (years), median (IQR)	25 (5)	26 (7)	0.564
Menarche (years), median (IQR)	12 (2)	12 (1)	0.730
Menstrual cycle (days), median (IQR)	30 (4)	30 (6)	0.895
Menstrual flow (days), median (IQR)	7 (1)	7 (3)	0.275
Education level, n (%)			
Tertiary	35 (97.2)	34 (94.4)	1.000
Postgraduate	I (2.8)	2 (5.6)	
Marital status, n (%)			
Single	29 (80.6)	30 (83.3)	1.000
Married	7 (19.4)	6 (16.7)	
Ethnicity, n (%)			
Malay	28 (77.8)	31 (86.1)	0.234
Chinese	5 (13.9)	4 (11.1)	
Indian	3 (8.3)	0 (0.0)	
Others	0 (0.0)	1 (2.8)	
Occupation, n (%)			
Unemployed	17 (47.2)	15 (42.9)	0.174
Employed	14 (38.9)	18 (51.4)	
Self-employed	I (2.8)	2 (5.7)	
Others	4 (11.1)	I (2.8)	

IQR: interquartile range.

study participants according to the treatment group. Continuous variables were reported as means and standard deviations, whereas categorical or ordinal variables were reported as absolute and relative frequencies. For outcome measurements, the analysis was based on intention-totreat, and participants with incomplete data were excluded. We calculated the mean scores for each life parameter and conducted an independent samples t-test to compare normally distributed data between the probiotic and placebo groups. We also used paired *t*-tests to compare baseline and post-treatment values within groups. A multivariate binary logistic regression model was run to assess the effect of potential confounders, such as the frequency of NSAID use and VAS, VRS, and SF12v2 scores that were included in the model as covariates for the treatment group. The results were further stratified according to confounding factors. Statistical significance was set at p < 0.05.

Results

Study demographics

The characteristics of the participants in the probiotic and placebo groups were similar (Table 1). Most women were single and had completed tertiary-level education. The age range of the participants was 23–41 years. The distribution of ethnic backgrounds was similar between the groups. The menstrual patterns were also similar in both groups.

A large proportion of the study participants was unemployed and comprised students.

Quality of life parameters

Before treatment, the quality of life scores in both groups was comparable. Pain score and severity were reported as moderate, while physical and mental health scores were above 50% (Table 2).

After 3 months of treatment, the probiotic group showed a significant improvement in mental health scores before and after treatment (mean change: 6.5, p=0.03) compared with the placebo group (mean change: 6.1, p=0.08). Both groups showed significant improvements in the VAS and VRS scores. The placebo group showed significantly lower VRS scores than the probiotic group. No other significant differences were found in other quality-of-life parameters.

The frequency of NSAID use was lower in the probiotic group (55.9%) than in the placebo group (64.7%); however, the difference was not statistically significant (Table 2). The odds ratio of using NSAIDs was 0.69 times lower for the probiotic group than for the placebo group (95% confidence interval: 0.26–1.83). Multivariate binary logistic regression analysis controlling for potential confounders, that is, frequency of NSAID use and VAS, VRS, and SF12v2 scores, demonstrated a significant association between the frequency of NSAID use and the severity score (VRS) (Table 3).

0.28

Quality of life	Mean (SD)	Probiotic (n=34)	Placebo (n=33)	Þª
Pain score (VAS)	Baseline	6.6 (1.7)	6.4 (1.7)	
	After treatment	4.0 (2.2)	3.9 (2.6)	0.69
	Changes	2.6 ^b (2.5)	2.5 ^b (3.1)	
Severity score (VRS)	Baseline	1.8 (0.7)	1.9 (0.7)	
	After treatment	1.3 (0.6)	1.1 (0.7)	0.03
	Changes	0.5 ^b (0.8)	0.8 ^b (0.9)	
Physical health score	Baseline	73.1 (16.9)	73.8 (14.0)	
(SF12v2)	After treatment	77.6 (13.4)	80.8 (18.4)	0.65
	Changes	4.5 (19.1)	6.9 (21.0)	
Mental health score	Baseline	68.2 (13.2)	69.2 (15.4)	
(SF12v2)	After treatment	74.7 (13.4)	75.4 (19.2)	0.74
	Changes	6.5 ^b (19.1)	6.1 (19.3)	
Frequency of NSAID use	Total	2.4 (3.6)	4.1 (6.6)	0.21

Table 2. Quality of life scores before and after treatment according to the treatment group (n=67).

SD: standard deviation; VAS: visual analog scale; VRS: verbal rating scale; SF12v2: Short-Form 12-item version 2; NSAID: nonsteroidal anti-inflammatory drug.

^aData were analyzed by independent *t*-test to determine differences between treatment groups.

Average per cycle

 $^{b}p < 0.05$, paired *t*-test analysis within the treatment group.

 Table 3. Multivariate logistic binary regression analysis of potential confounders showing the association between VRS score and frequency of NSAID use.

Co-variates	AOR	95% CI	Þ
Difference in VAS	0.776	0.58-1.03	0.23
Difference in physical health score	0.973	0.93-1.02	0.53
Difference in mental health score	0.983	0.93-1.04	0.08
Frequency of NSAID use	1.39	1.11–1.73	0.004

AOR: adjusted odd ratio; CI: confidence interval; NSAID: nonsteroidal anti-inflammatory drug; VAS: visual analog scale.

Stratification analysis

(number of tablets)

The results were further stratified according to NSAID use. The placebo group showed significantly better VRS scores when NSAIDs were administered. No significant differences were noted between the groups with respect to other parameters. In addition, the significance of improvement in mental health scores in the probiotic group was lost after adjusting for NSAID use (Table 4).

Inflammatory markers

No significant difference was found in the levels of inflammatory markers before and after treatment in either group; however, the sample size for this analysis was small owing to inadequate funding (Figure 2).

Side effects of treatment

A few side effects such as diarrhea, bloating, and fever were reported in the placebo group (n=3) but no side effects were reported in the probiotic group.

Discussion

0.8 (1.2)

The aim of this trial was to assess the effects of a 3-month oral probiotic supplementation in women with primary dysmenorrhea compared with placebo on the quality of life and the effect on inflammatory cytokines on the same. Multiple tools were used to assess the quality of life, including subjective (VAS, VRS, and SF12v2 questionnaires) and objective (frequency of NSAID use) measures. Regarding inflammatory cytokines, we chose IL-6, IL-8, and TNF- α , since their correlations with primary dysmenorrhea are well established in the literature.^{9,11}

1.3 (2.2)

Oral supplementation with the tested probiotics for women with primary dysmenorrhea did not significantly improve their quality of life or inflammatory markers compared with placebo. However, there was no significant reduction in NSAID use in the probiotic group either. Long-term NSAID use is associated with side effects, such as renal impairment and peptic ulcer disease; hence, reduced dependence on NSAIDs could help reduce the risk of side effects, thus improving the quality of life.

The most likely explanation for the lack of differences found between the probiotic and placebo groups was the confounding effect of NSAIDs, which could mask any improvement attributed to probiotics. The highest pain score improvement was observed in the probiotic subgroup group that did not use NSAIDs (Table 4). However, as the sample size was small, the reduction was not significantly different from that in the placebo group. In addition, as evidenced by the better scores among NSAID users especially in the placebo group as compared with the probiotic group, the effect of probiotics, if any, was less significant than that of NSAIDs. A study comparing the effect of probiotics alone versus NSAIDs would provide a clearer

Quality of life	Mean (SD)	Did not take NSAIDs		Took NSAIDs			
		Probiotic $(n = 15)$	Placebo ($n = 12$)	Þª	Probiotic $(n = 19)$	Placebo (n=22)	Þª
Pain score (VAS)	Changes	-2.8 (2.6)	-2.6 (2.8)	0.84	-2.5 (2.4)	-2.4 (3.3)	0.95
Severity score (VRS)	Changes	-0.3 (0.52)	-0.5 (0.58)	0.58	-0.4 (0.8)	-1.0 (1.0)	0.03
Physical health score	Changes	4.0 (18.6)	11.3 (20.2)	0.34	4.8 (25.6)	4.4 (21.6)	0.96
Mental health score	Changes	8.4 (23.6)	9.9 (21.9)	0.87	7.2 (15.2)	4.0 (17.9)	0.55

Table 4. Changes in quality of life scores after stratification by NSAID use.

SD: standard deviation; NSAID: nonsteroidal anti-inflammatory drug; VAS: visual analog scale; VRS: verbal rating scale. ^aUsing nonparametric independent samples Mann–Whitney U test.



Figure 2. Boxplot showing concentration of inflammatory markers (IL-6, IL-8, and TNF- α) at baseline and after treatment comparing between treatment groups (placebo, n=6; probiotic, n=8). p > 0.05, using independent Mann–Whitney U test.

picture, but this is not ethically possible at present because NSAIDs are the current standard treatment for dysmenorrhea and the pain can be severe; therefore, inhibiting access to NSAIDs is deemed unethical. Regarding inflammatory cytokines, it is possible that the differences in concentration between baseline and post-treatment levels were minute, especially when the samples were derived from the serum rather than the local endometrial or vaginal fluid. Ideally, a study using a highly sensitive immunoassay and larger sample size can detect minute changes; however, this study could not achieve this owing to budget constraints.

This double-blind randomized-controlled trial compared the effects of oral probiotics and placebo in women with primary dysmenorrhea. The double-blind design reduced observer bias, and the randomization process ensured similarity in baseline characteristics between the groups. Three months of supplementation was thought to be sufficient to produce an effect over three menstrual cycles, as this duration is usually prescribed for oral contraceptive pills or NSAIDs. The broad inclusion criteria enhanced external validity, and low exclusion (n=6/78) and dropout rates (n=11/72) ensured internal validity. The use of an intention-to-treat analysis further minimized the risk of bias and retained the randomization effect. Nevertheless, the impact on the intestinal microbiota was not tested, which is one of the limitations of this trial. In healthy patients, oral probiotics did not significantly affect fecal microbiota diversity.²³ Other studies have shown that oral probiotic supplementation during pregnancy promoted a healthy vaginal microbiota that suppressed preterm labor.²⁴ Probiotics have also been shown to positively modulate female microbiota and alleviate endocrine and fertility-related disorders.²⁵

To our knowledge, this is the first study to assess the role of probiotics in primary dysmenorrhea. Previous trials have demonstrated the beneficial effect of oral probiotics in reducing dysmenorrhea and the severity of symptoms in

patients with endometriosis. In a randomized-controlled trial that investigated the effect of Lactobacillus gasseri GG on endometriosis symptoms, menstrual pain was significantly reduced in the Lactobacillus group compared with the placebo.¹⁷ Another pilot randomized-controlled trial of 37 women with endometriosis also reported significant improvement in pain score after use of oral Lactobacillus LactoFem for 8 weeks compared with placebo.¹⁸ However, the effects of probiotics have not been tested in women with primary dysmenorrhea. Many previous trials on primary dysmenorrhea have used alternative therapies, such as herbal remedies, behavioral interventions, exercise, and acupuncture, which showed improvement in menstrual pain; however, many of the studies had unclear protocols or low methodological quality; hence, their results were inconclusive.²⁶⁻²⁸ Their assessments were also largely based on pain scores, and many did not assess the quality of life using specific questionnaires, such as those used in this study.²⁹⁻³¹ Other studies that assessed the quality of life of patients with primary dysmenorrhea were limited to observational studies and used different questionnaires.¹⁻³ A questionnaire that is sensitive enough to assess conditions like primary dysmenorrhea is yet to be decided.

Previous studies have identified the possible mechanisms by which oral probiotics may improve inflammatory conditions. van Baarlen et al.¹⁶ showed that the colonization of the gastrointestinal mucosa by probiotic bacteria altered the expression of many genes involved in mucosal immunity and correlated significantly with the processes involved in regulating immune responses. In another study, Lactobacillus rhamnosus was reported to reduce proinflammatory TNF- α production in gram-positive bacteria, as evidenced by reduced level of C-reactive protein.³² In women with polycystic ovarian syndrome, oral probiotic supplementation has been shown to reduce the levels of inflammatory markers such as IL-6, IL-10, C-reactive protein, and TNF-α.³³ A similar positive effect was also seen in patients with major depressive disorder, where oral probiotic supplementation led to a reduction in the urinary cortisol levels and significant improvement in depressive symptoms.³⁴ Other studies have shown that probiotics can affect metabolism and modulate inflammatory responses in female reproductive processes. A systematic review of the effects of probiotics in pregnancy showed a positive effect of probiotics on maternal metabolism, such as a reduction in C-reactive protein, fasting glucose, and the rate of preeclampsia.³⁵ Another large prospective cohort study also found a reduction in the rates of preterm delivery and preeclampsia in women who were administered probiotic milk during late pregnancy.³⁶

Therefore, probiotic supplementation has a positive effect on inflammatory conditions.

There is growing evidence on the association between gut dysbiosis, increased estrogen reabsorption, and dysmenorrhea. Gut dysbiosis refers to any perturbation in the diversity and/or number of healthy populations of intestinal microbiota that can occur through diet, age, ethnicity, medication, smoking, and alcohol intake.³⁷ Gut dysbiosis affects beta-glucuronidase enzyme activity, which is the main deconjugation pathway of estrogen in the intestinal microbiota. This, in turn, affects estrogen metabolism and exacerbates estrogen-driven conditions like endometriosis, which is another major cause of dysmenorrhea.³⁸ The intestinal microbiota regulate estrogen metabolism via the estrogen–gut axis and estrobolome, and dysfunctions in these mechanisms can cause gynecological conditions such as endometriosis, infertility, chronic pelvic pain, and dysmenorrhea.^{37,39}

In addition, endometriosis has been associated with a deficiency in *Lactobacillus* and proliferation of gram-negative organisms in many animal and human studies.^{39–41} Given the potential of probiotic supplementation to reverse gut dysbiosis and restore estrogen homeostasis, oral probiotics can be used to treat gynecological conditions, especially disorders that have an inflammatory origin and are estrogen-driven, like primary dysmenorrhea.

However, this study had some limitations. The sample size was small; hence, it was underpowered to detect a significant difference in outcomes such as inflammatory markers and the frequency of NSAID use. The confounding effects of NSAID use were not considered. Second, although the compliance rate was reported to be >90% in each group, this was done via self-reporting, which is prone to reporter bias. We also did not measure the evolution of quality-of-life parameters per month and measured them only at two time points, before and after the 3-month treatment. Any benefit observed in the first 2 months could have been missed or lost at the end of treatment as a result of a lack of compliance. Furthermore, as pain in primary dysmenorrhea typically occurs more than 24 to 72 h, the SF12v2 questionnaire, which inquired about one's experience over a monthly period, might not be able to capture the impact of pain during these particular days.

Given the increasing evidence on the benefits of probiotics in reproductive and general health conditions, our study aimed to assess the effect of probiotics on primary dysmenorrhea, a common ailment affecting women of reproductive age that constitutes a major health and economic burden owing to its impact on workplace/school productivity. A previous metabolomic study of primary dysmenorrhea revealed a reduction in progesterone levels and several metabolomic conditions that could explain the mechanism of increased PG expression and pain perception.⁴² Current strategies rely on the use of analgesics, mainly NSAIDs, which offer symptomatic relief but do not target the underlying pathogenesis. Probiotics rich in Lactobacillus promote a healthy gut environment and have been proven to be effective against inflammatory diseases such as eczema, irritable bowel syndrome, and endometriosis.^{15,25,43} These benefits of probiotics may be extrapolated to primary dysmenorrhea since this condition is also inflammatory in origin, although its exact pathogenesis is still unclear.

Conclusion

In conclusion, the tested oral probiotics did not affect the quality of life or levels of inflammatory cytokines in women with primary dysmenorrhea. However, the probiotics showed the potential to reduce NSAID use and improve pain and mental health; however, these results need to be further elucidated in larger trials, preferably using non-NSAID analgesics such as paracetamol or tramadol.

Declarations

Ethics approval and consent to participate

The study was conducted in accordance with the Declaration of Helsinki, approved by the Institutional Ethics Committee of the National University of Malaysia (FF-2018-204, 5 June 2018), and registered at ClinicalTrials.gov (NCT04119011). Written informed consent was obtained from all participants involved in the study.

Consent for publication

All subjects involved in the study consented for publication of their data.

Author contribution(s)

Izyan Atiqah Zakaria: Data curation; Formal analysis; Project administration; Visualization; Writing—original draft; Writing—review & editing.

Nur Aini Mohammed Zain: Conceptualization; Data curation; Project administration; Writing—original draft.

Chew Kah Teik: Conceptualization; Methodology; Resources; Writing—original draft.

Muhammad Azrai Abu: Conceptualization; Investigation; Methodology; Writing—original draft.

Ani Amelia Zainuddin: Conceptualization; Funding acquisition; Resources; Writing—review & editing.

Nor Haslinda Abdul Aziz: Conceptualization; Investigation; Methodology; Resources; Supervision; Writing—original draft.

Nazarudin Safian: Formal analysis; Software; Validation; Writing—review & editing.

Norfilza Mohd Mokhtar: Conceptualization; Investigation; Resources; Writing—review & editing.

Raja Affendi Raja Ali: Conceptualization; Funding acquisition; Methodology; Resources; Writing—review & editing.

Ng Beng Kwang: Investigation; Resources; Writing—review & editing.

Nor Azlin Mohamed Ismail: Investigation; Resources; Supervision; Writing—review & editing.

Muhammad Rafiuddin Hamizan: Formal analysis; Validation; Writing—original draft.

Wira Sorfan Ab Razak: Methodology; Software; Writing—original draft.

Abdul Ghani Nur Azurah: Conceptualization; Funding acquisition; Methodology; Resources; Supervision; Visualization; Writing—review & editing.

Acknowledgements

The authors would like to thank all the clinical and supporting staff of the Department of Obstetrics and Gynecology, Hospital Canselor Tunku Muhriz, for their support and involvement throughout the study.

Funding

The author(s) disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This research was funded by the Fundamental Fund of the National University of Malaysia (grant no. FF-2018-204). We also received sponsorship from B-crobes Sdn Bhd for a 3-month supply of placebo and probiotic sachets for all 72 participants.

Competing interests

The author(s) declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Availability of data and materials

The data sets used in this study can be found in the supplementary files submitted along with the article. Please contact the corresponding authors for further details.

Disclosure

This article in its earlier version has been deposited in a preprint server Authorea on 16 October 2020 with DOI: 10.22541/au.160288198.88055969/v1, website link: https://www.authorea.com/users/367751/articles/487061-the-effect-of-probiotics-on-quality-of-life-in-women-with-primary-dysmenorrhoea-a-randomized-double-blind-placebo-controlled-trial

ORCID iD

Izyan Atiqah Zakaria (D) https://orcid.org/0000-0003-3816-7589

Supplemental material

Supplemental material for this article is available online.

References

- Nur Azurah AG, Sanci L, Moore E, et al. The quality of life of adolescents with menstrual problems. *J Pediatr Adolesc Gynecol* 2013; 26: 102–108.
- Fernández-Martínez E, Onieva-Zafra MD and Parra-Fernández ML. The impact of dysmenorrhea on quality of life among Spanish female university students. *Int J Environ Res Public Health* 2019; 16: 713.

- Hashim RT, Alkhalifah SS, Alsalman AA, et al. Prevalence of primary dysmenorrhea and its effect on the quality of life amongst female medical students at King Saud University, Riyadh, Saudi Arabia. *Saudi Med J* 2020; 41(3): 283–289.
- Iacovides S, Avidon I and Baker FC. What we know about primary dysmenorrhea today: a critical review. *Hum Reprod Update* 2015; 21(6): 762–778.
- Itani R, Soubra L, Karout S, et al. Primary dysmenorrhea: pathophysiology, diagnosis, and treatment updates. *Korean J Fam Med* 2022; 43(2): 101–108.
- 6. Proctor M and Farquhar C. Diagnosis and management of dysmenorrhoea. *BMJ* 2006; 332: 1134–1138.
- Harel Z. Dysmenorrhea in adolescents and young adults: from pathophysiology to pharmacological treatments and management strategies. *Expert Opin Pharmacother* 2008; 9(15): 2661–2672.
- Bieglmayer C, Hofer G, Kainz C, et al. Concentrations of various arachidonic acid metabolites in menstrual fluid are associated with menstrual pain and are influenced by hormonal contraceptives. *Gynecol Endocrinol* 1995; 9(4): 307–312.
- Ma H, Hong M, Duan J, et al. Altered cytokine gene expression in peripheral blood monocytes across the menstrual cycle in primary dysmenorrhea: a case-control study. *PLoS ONE* 2013; 8(2): e55200.
- Dawood MY. Primary dysmenorrhea. Obstet Gynecol 2006; 108: 428–441.
- 11. Barcikowska Z, Rajkowska-Labon E, Grzybowska ME, et al. Inflammatory markers in dysmenorrhea and therapeutic options. *Int J Environ Res Public Health* 2020; 17: 1191.
- Barańska A, Błaszczuk A, Kanadys W, et al. Oral contraceptive use and breast cancer risk assessment: a systematic review and meta-analysis of case-control studies, 2009– 2020. *Cancers (Basel)* 2021; 13: 5654.
- Bindu S, Mazumder S and Bandyopadhyay U. Nonsteroidal anti-inflammatory drugs (NSAIDs) and organ damage: a current perspective. *Biochem Pharmacol* 2020; 180: 114147.
- Oladosu FA, Tu FF and Hellman KM. Nonsteroidal antiinflammatory drug resistance in dysmenorrhea: epidemiology, causes, and treatment. *Am J Obstet Gynecol* 2018; 218(4): 390–400.
- Hemarajata P and Versalovic J. Effects of probiotics on gut microbiota: mechanisms of intestinal immunomodulation and neuromodulation. *Therap Adv Gastroenterol* 2013; 6(1): 39–51.
- van Baarlen P, Troost F, van der Meer C, et al. Human mucosal in vivo transcriptome responses to three lactobacilli indicate how probiotics may modulate human cellular pathways. *Proc Natl Acad Sci USA* 2011; 108(suppl. 1): 4562–4569.
- Itoh H, Uchida M, Sashihara T, et al. Lactobacillus gasseri OLL2809 is effective especially on the menstrual pain and dysmenorrhea in endometriosis patients: randomized, double-blind, placebo-controlled study. *Cytotechnology* 2011; 63(2): 153–161.
- Khodaverdi S, Mohammadbeigi R, Khaledi M, et al. Beneficial effects of oral lactobacillus on pain severity in women suffering from endometriosis: a pilot placebo-controlled randomized clinical trial. *Int J Fertil Steril* 2019; 13(3): 178–183.

- Safdari F and Parvin N. The effect of mefenamic acid and melissa officinalis on primary dysmenorrhea: a randomized clinical trial study. *Int J Pharmacogn Phytochem Res* 2016; 8: 1286–1292.
- 20. Ware JE, Kosinski MA, Turner-Bowker DM, et al. *How to score: version 2 of the SF-12v2 health survey*, vol. 22. Boston, MA: Health Assessment Lab, 2002, pp. 32–35.
- Hayes CJ, Bhandari NR, Kathe N, et al. Reliability and validity of the medical outcomes study short form-12 version 2 (SF-12v2) in adults with non-cancer pain. *Healthcare* (*Basel*) 2017; 5(2): 22.
- 22. Hjermstad MJ, Fayers PM, Haugen DF, et al. Studies comparing Numerical Rating Scales, Verbal Rating Scales, and Visual Analogue Scales for assessment of pain intensity in adults: a systematic literature review. *J Pain Symptom Manage* 2011; 41(6): 1073–1093.
- Noh C-K, Kim BS, Hong G, et al. Effects of the administration of probiotics on fecal microbiota diversity and composition in healthy individuals. *J Neurogastroenterol Motil* 2018; 24: 452–459.
- 24. Kirihara N, Kamitomo M, Tabira T, et al. Effect of probiotics on perinatal outcome in patients at high risk of preterm birth. *J Obstet Gynaecol Res* 2018; 44(2): 241–247.
- López-Moreno A and Aguilera M. Probiotics dietary supplementation for modulating endocrine and fertility microbiota dysbiosis. *Nutrients* 2020; 12(3): 75.
- Cho SH and Hwang EW. Acupuncture for primary dysmenorrhoea: a systematic review. *BJOG* 2010; 117(5): 509–521.
- Pellow J and Nienhuis C. Medicinal plants for primary dysmenorrhoea: a systematic review. *Complement Ther Med* 2018; 37: 13–26.
- 28. Armour M, Ee CC, Naidoo D, et al. Exercise for dysmenorrhoea. *Cochrane Database Syst Rev* 2019; 9(9): CD004142
- 29. Igwea SE, Tabansi-Ochuogu CS and Abaraogu UO. TENS and heat therapy for pain relief and quality of life improvement in individuals with primary dysmenorrhea: a systematic review. *Complement Ther Clin Pract* 2016; 24: 86–91.
- McGovern CE and Cheung C. Yoga and quality of life in women with primary dysmenorrhea: a systematic review. J Midwifery Womens Health 2018; 63(4): 470–482.
- 31. Jenabi E. The effect of ginger for relieving of primary dysmenorrhoea. *J Pak Med Assoc* 2013; 63(1): 8–10.
- Zhang L, Xu YQ, Liu HY, et al. Evaluation of *Lactobacillus rhamnosus* GG using an *Escherichia coli* K88 model of piglet diarrhoea: effects on diarrhoea incidence, faecal microflora and immune responses. *Vet Microbiol* 2010; 141: 142–148.
- Ghanei N, Rezaei N, Amiri GA, et al. The probiotic supplementation reduced inflammation in polycystic ovary syndrome: a randomized, double-blind, placebo-controlled trial. *J Funct Foods* 2018; 42: 306–311.
- Kazemi A, Noorbala AA, Azam K, et al. Effect of prebiotic and probiotic supplementation on circulating pro-inflammatory cytokines and urinary cortisol levels in patients with major depressive disorder: a double-blind, placebo-controlled randomized clinical trial. *J Funct Foods* 2019; 52: 596–602.
- Lindsay KL, Walsh CA, Brennan L, et al. Probiotics in pregnancy and maternal outcomes: a systematic review. J Matern Fetal Neonatal Med 2013; 26(8): 772–778.

- 36. Nordqvist M, Jacobsson B, Brantsæter A-L, et al. Timing of probiotic milk consumption during pregnancy and effects on the incidence of preeclampsia and preterm delivery: a prospective observational cohort study in Norway. *BMJ*
- Open 2018; 8: e018021.
 37. Elkafas H, Wall M, Al-Hendy A, et al. Gut and genital tract microbiomes: dysbiosis and link to gynecological disorders. *Front Cell Infect Microbiol* 2022; 12: 1059825.
- Baker JM, Al-Nakkash L and Herbst-Kralovetz MM. Estrogen-gut microbiome axis: physiological and clinical implications. *Maturitas* 2017; 103: 45–53.
- Salliss ME, Farland LV, Mahnert ND, et al. The role of gut and genital microbiota and the estrobolome in endometriosis, infertility and chronic pelvic pain. *Hum Reprod Update* 2021; 28: 92–131.

- 40. Bailey MT and Coe CL. Endometriosis is associated with an altered profile of intestinal microflora in female rhesus monkeys. *Hum Reprod* 2002; 17(7): 1704–1708.
- Leonardi M, Hicks C, El-Assaad F, et al. Endometriosis and the microbiome: a systematic review. *BJOG* 2020; 127(2): 239–249.
- 42. Fang L, Gu C, Liu X, et al. Metabolomics study on primary dysmenorrhea patients during the luteal regression stage based on ultra performance liquid chromatography coupled with quadrupole-time-of-flight mass spectrometry. *Mol Med Rep* 2017; 15(3): 1043–1050.
- Russo R, Edu A and De Seta F. Study on the effects of an oral lactobacilli and lactoferrin complex in women with intermediate vaginal microbiota. *Arch Gynecol Obstet* 2018; 298(1): 139–145.